Author’s response to reviews

Title: The effects of psychological interventions on depression and anxiety among Chinese adults with cancer: a meta-analysis of randomized controlled studies

Authors:

Yi-Long Yang (yyyll390@163.com)
Guo-Yuan Sui (lgc0519@126.com)
Guang-Cong Liu (584883734@qq.com)
De-Sheng Huang (dshuang@hotmail.com)
Si-Meng Wang (wsmns2002@aliyun.com)
Lie Wang (liewang@mail.cmu.edu.cn)

Version: 2 Date: 10 November 2014

Author’s response to reviews:

Dear Sir or Madam,

We have revised the manuscript by Yi-Long Yang et al. entitled “The effects of psychological interventions on depression and anxiety among Chinese adults with cancer: a meta-analysis of randomized controlled studies”. We follow the editor advice and addressed all of these concerns of the three reviewers. According to the reviewers’ question and advice, we have provided a point-by-point description of the changes made as follows.